Author’s response to reviews

Title: Evaluation of the Physician Quality Improvement Initiative: The expected and unexpected opportunities

Authors:
Kirsten Wentlandt (Kirsten.Wentlandt@uhn.ca)
Andrea Bracaglia (andream.bracaglia@gmail.com)
James Drummond (jamesdrummond50@gmail.com)
Lindsay Handren (ldhandren@gmail.com)
Joshua McCann (joshuajeremiahmccann@gmail.com)
Catherine Clarke (Catherine.Clarke@uhn.ca)
Niki Degendorfer (Niki.Degendorfer@uhn.ca)
Charles Chan (Charlie.Chan@uhn.ca)

Version: 2 Date: 03 Dec 2015

Author’s response to reviews:

December 3, 2015

Editor-in-Chief

BMC Medical Education

We would like to thank you for reviewing our paper entitled “Evaluation of the Physician Quality Improvement Initiative: The expected and unexpected opportunities” for consideration for publication in the journal BMC Medical Education.
We have reviewed the editor’s comments and outlined our respective revisions and comments.

1. Please provide the full name and affiliation of the ethics committee that reviewed this study.

The full name/affiliation of the ethics committee has been added to page 6 (Methods section) and page 21 (under Ethics Approval). Please see highlighted area on both pages.

2. add all authors' email-addresses to the title page.

All of the author’s email addresses have been added to the Title page.

3. Please format the title page according to the instructions for authors

This has been done.

4. Please include CI/AC/Ack at the end of the ms.

The Contributions, Funding, Acknowledgement, Conflicts of Interest, and Ethics approval outline has been moved to the end of the document on page 21.

Thank you in advance for your consideration of our paper for publication in the journal BMC Medical Education.

Sincerely,

Kirsten Wentlandt PhD MHSci MD CCFP
Assistant Professor, University of Toronto
Staff Physician, Department of Psychosocial Oncology and Palliative Care
UHN- Toronto General Hospital
9NU- 925, 200 Elizabeth St
Toronto, ON M5G 2C4
Tel. 416 340 4800 ext 5247
Fax 416 340 4823